Author's response to reviews

Title: Local versus General Anesthesia for Transcatheter Aortic Valve Implantation (TAVI) - Systematic Review and Meta-Analysis

Authors:

Georg M Fröhlich (georg.froehlich@gmx.at)
Alexandra J Lansky (alexandra.lansky@yale.edu)
John Webb (john.webb@vch.ca)
Marko Roffi (marco.roffi@huge.ch)
Stephan Toggweiler (stefan.toggweiler@cevi.ch)
Markus Reinthaler (markusreinthaler@gmx.at)
Duolao Wang (duolao.wang@lshtm.ac.uk)
Nevil Hutchinson (nevilhutchinson@nhs.net)
Olaf Wendler (olaf.wendler@nhs.net)
David Hildick-Smith (david.hildick-smith@bsuh.nhs.uk)
Pascal Meier (pascalmeier74@gmail.com)

Version: 2
Date: 20 January 2014

Author's response to reviews:

Please find attached our point-by-point reply, we appreciate the careful appraisal of our paper.